Author's response to reviews

Title: Percutaneous treatment of spontaneous left main coronary artery dissection using drug-eluting stent.

Authors:

Christos Graidis (dimitriadis1980@gmail.com)
Dimokritos Dimitriadis (dimitriadis.dimokritos@gmail.com)
Vasileios Karasavvidis (v.karasavvides@gmail.com)
Georgios Dimitriadis (georgiosdim@yahoo.gr)
Efstathia Argyropoulou (drteti@yahoo.gr)
Fotios Oikonomou (economouf@gmail.com)
George Spiromitros (dim_gio@hotmail.com)
Antonios Antoniou (dim_gio@hotmail.com)
George Karakostas (geor-kar@hotmail.com)

Version: 3 Date: 12 November 2014

Author's response to reviews:

Dear reviewers all the necessary changes have been made according to your comments. We hope that you will be satisfied and honour us with your approval for publication!

Yours sincerely Christos Graidis